Author's response to reviews

Title: Post traumatic stress symptoms and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study

Authors:

Shirley Telles (shirleytelles@gmail.com)
Nilkamal Singh (naosekpamkamal@gmail.com)
Meesha Joshi (meeshie26@gmail.com)
Acharya Balkrishna (pyp@divyayoga.com)

Version: 14 Date: 25 February 2010

Author's response to reviews:

There was a typing error in the table section i.e., in Table 1, row no 4, the value of years of education for the yoga group was typed as 6.4.1 ± 1.1 we have changed this to 6.4 ± 1.1 which is the correct value and in the same row the value of years of education for the control group was typed as 5.9± 2.0 we have changed this to 5.9 ± 2.0

We regret for this change in this stage.